Author's response to reviews

Title: Quadricuspid aortic valve by using intraoperative transesophageal echocardiography

Authors:

Zhenghua Xiao MD (lg886agl@126.com)
Wei Meng MD,PHD (meng_wei_1111@yahoo.com)
Eryong Zhang MD,Prof (xzh8585@126.com)

Version: 3 Date: 28 August 2010

Author's response to reviews: see over
Response to reviewer

Major revisions  (we require the author to make these changes)

Abstract: please ensure that the Abstract matches exactly what is in the submission system

I have ensured that the Abstract matched exactly what was in the submission system

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Title: please use sentence case in the title, both in the manuscript file and the submission system. Capitalise only the first word and proper nouns, and do not include full stops/periods. For example, 'The Title of My Prestigious Publication: A Scientific Study.' should be changed to 'The title of my prestigious publication: a scientific study'

Author List: Please put a space between Christian names and surnames in the author list.

Qualifications: Please remove the authors’ qualifications (e.g. PhD / MD) or job titles from the submissions system

Highlighting: Please remove the highlighting from figure 3.

Figure legends: please move from page 3 to the end of the manuscript.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

I have modified all the Minor revisions.

Thank you very much! Best wishes!